Author’s response to reviews

Title: Stability and change in screen-based sedentary behaviours and associated factors among Norwegian children in the transition between childhood and adolescence

Authors:

Mekdes Gebremariam (mekdes.gebremariam@medisin.uio.no)
Torunn Totland (t.h.totland@medisin.uio.no)
Lene Frost Andersen (l.f.andersen@medisin.uio.no)
Ingunn Bergh (ingunn.holden.bergh@medisin.uio.no)
Mona Bjelland (mona.bjelland@medisin.uio.no)
May Grydeland (may.grydeland@medisin.uio.no)
Ommundsen Yngvar (yngvar.ommundsen@nih.no)
Nanna Lien (nanna.lien@medisin.uio.no)

Version: 4 Date: 22 January 2012

Author’s response to reviews:

Dear editor,

We would like for this version of the revised manuscript to be considered instead of the one we previously uploaded due to small changes made. All changes have been noted with track changes. The changes made are minor and in line with the revisions described in the cover letter with replies to reviewers already uploaded (Reply to reviewers doc. above). Please use that cover letter, since no changes to that cover letter are needed.

Gebremariam MK,
Corresponding author.